Author's response to reviews

Title: Dynamics of C-reactive protein and white blood cell count in critically ill patients with nosocomial Gram positive vs. Gram negative bacteremia: a historical cohort study

Authors:

Dominique M Vandijck (Dominique.Vandijck@UGent.be)
Eric A Hoste (Eric.Hoste@UGent.be)
Stijn I Blot (Stijn.Blot@UGent.be)
Pieter O Depuydt (Pieter.Depuydt@UGent.be)
Renaat A Peleman (Renaat.Peleman@UGent.be)
Johan M Decruyenaere (Johan.Decruyenaere@UGent.be)

Version: 6 Date: 10 September 2007

Author's response to reviews:

Dear Editor

We hereby submit the revised version (final version) of our manuscript entitled 'Dynamics of c-reactive protein and white blood cell count in critically ill patients with nosocomial gram-positive vs. gram-negative bacteremia: a historical cohort study' (MS 1336187211956803) for publication in BMC Infectious Diseases.

In meantime, we remain,

Most sincerely,

Dominique Vandijck